Author's response to reviews

Title: The Mayer-Rokitansky-Kuster-Hauser syndrome (congenital absence of uterus and vagina) - phenotypic manifestations and genetic approaches.

Authors:

Daniel Guerrier (daniel.guerrier@univ-rennes1.fr)
Thomas Mouchel (thomas.mouchel@club-internet.fr)
Laurent Pasquier (laurent.pasquier@chu-rennes.fr)
Isabelle Pellerin (isabelle.pellerin@univ-rennes1.fr)

Version: 3 Date: 13 September 2005

Author’s response to reviews: see over
To: Reviewers of the manuscript: The Mayer-Rokitansky-Küster-Hauser syndrome (congenital absence of uterus and vagina) - phenotypic manifestations and genetic approaches. by D. GUERRIER et al.

Dear Sir, Madam,

Please find a revised version of the manuscript cited above. As requested, the whole text was amended by an English native, Dr. D.W. Dresser whose work is acknowledged. I do not give detail on the corrections as they were made throughout the whole manuscript (grammar and style).

In addition I updated the "Particular case of Wnt genes" section as an article was very recently publish by Clement-Ziza et al. (reference # 91) who investigated the Wnt4 gene in several MRKH patients. This only reinforces what I previously wrote.

I hope this new version will now be suitable for publication in the Journal of Negative Results in BioMedicine.

Best regards,

D. Guerrier, corresponding author